Author's response to reviews

Title: Subannular left ventricular pseudoaneurysm following mitral valve replacement

Authors:

Narayanan Namboodiri (kknamboodiri@yahoo.co.in)
Santosh K Dora (kskdora@hotmail.com)
Bejoy Thomas (bejoy@sctimst.ac.in)
Manoranjan Mishra (mishra@sctimst.ac.in)

Version: 4 Date: 29 April 2008

Author's response to reviews:

Dear Sir,

Submitting the revision of article "Subannular left ventricular pseudoaneurysm following mitral valve replacement", MS : 9095289661445842 for Journal of Cardiothoracic Surgery

The following changes have been made as per instruction.
Format changes requested
-----------------------------------------------
Please remove 'Type of manuscript: case report' from the title page.- done
Please remove the short title, key words and word count.- done

Affiliations- Each affiliation should be written in full, in the following format: department, institute, city, and country.
This may mean repeating institutional addresses if the authors are from different departments within an institution.
Can we also ask you to link each author to his or her corresponding affiliation with the use of superscript numbers?
done

Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system -done

Please remove 'Manuscript' from the top of page 2.- done

Authors’ contributions - Please place the Authors’ contributions section towards the end of the manuscript text, directly preceding the Acknowledgments. -done
Authors’ contributions- please use the following format (please use initials to refer to each author’s contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

Acknowledgments -If you have any acknowledgments, please place an Acknowledgments section at the end of the manuscript text, directly preceding the Reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgments.

Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgments section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.

Please check the author details for reference 4 are correct.

Please change Figure1 and Figure2 to Figure 1 and Figure 2 accordingly.

References - References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text
done as per the direction

Thank you

Namboodiri N